Author's response to reviews

**Title:** Prevalence of anemia and deficiency of iron, folic acid, and zinc in children under 2 years of age and beneficiaries of the Mexican Social Security Institute.

**Authors:**

Ximena Duque (ximena.duque@imss.gob.mx)
Sergio Flores (sergio.flores@imss.gob.mx)
Samuel Flores (floreshuertamd@yahoo.com)
Ignacio Mendez (imendez@servidor.unam.mx)
Sergio Munoz (munozs@ufro.cl)
Bernardo Turnbull (bernardo.turnbull@imss.gob.mx)
Gloria Martinez (gloria.martineza@imss.gob.mx)
Rosa I Ramos (rosa.ramosh@imss.gob.mx)
Marco A Gonzalez (marco.unzaga@imss.gob.mx)
Maria-Eugenia Mendoza (mendoza_ma13@hotmail.com)
Homero Martinez (homero@rand.org)

**Version:** 4  **Date:** 20 November 2007

**Author's response to reviews:**

We corrected all the typographical errors marked by reviewer 2.

In addition, we attended the request of the editor about the institutional review board that granted ethical approval for the study. We incorporated these details in the Methods section Page 6 line 4. "Regarding ethical considerations, the Institutional Research Review Board approved the protocol study and, the informed consent letter was signed by mothers of the children in the subsample."

We also followed the formatting checklist.